Author's response to reviews

Title: Expression-based Pathway Signature Analysis (EPSA): Mining publicly available microarray data for insight into human disease

Authors:

Jessica D Tenenbaum (jessie.tenenbaum@duke.edu)
Michael G Walker (mwalker@walkerbioscience.com)
Paul J Utz (piutz@stanford.edu)
Atul J Butte (abutte@stanford.edu)

Version: 5 Date: 6 October 2008

Author's response to reviews: see over
October 10, 2008

Melissa Norton, MD
Editor-in-Chief, BMC Medical Genomics

Dear Dr. Norton (and other editors):

This cover letter accompanies the final proof of my manuscript entitled Expression-based Pathway Signature Analysis (EPSA): Mining publicly available microarray data for insight into human disease, manuscript number 7411472591952180.

Changes in this version include only minor modifications:

- Removal of capitalizations
- Insertion of missing words, e.g. a, an, to
- Deletion of duplicate words
- Insertion of missing periods
- On page 5, we replaced “this step” with “Extracting this subset of genes” for clarity
- Subject/verb matching

Thank you for the opportunity to share our work with your readers.

Sincerely,

Jessica Tenenbaum, PhD
Project Leader
Duke Translational Medicine Institute
PO Box 17969
Durham, NC 27615
jessie.tenenbaum@duke.edu
+1 919-668-8811